

Trends in aggregate employment, hours worked per worker, and the long-run labor wedge

Epstein, Brendan and Mukherjee, Rahul and Finkelstein Shapiro, Alan and Ramnath, Shanthi

March 2020

Online at https://mpra.ub.uni-muenchen.de/99289/ MPRA Paper No. 99289, posted 30 Mar 2020 09:15 UTC

Trends in Aggregate Employment, Hours Worked per Worker, and the Long-Run Labor Wedge^{*}

Brendan Epstein[†] Alan Finkelstein Shapiro[‡] Rahul Mukherjee[§] Shanthi Ramnath[¶]

Abstract

Hours worked are fundamentally important for aggregate economic activity, yet canonical macroeconomic models fail dramatically at tracking its long run trends. We develop an intuitive and tractable extension of the canonical model that decomposes trend hours into extensive and intensive margins via household-side employment-attainment costs and firm-side employment adjustment costs. Its predictions track very well the trend behavior of hours, and its two underlying margins, in the United States and a host of OECD countries. Our framework is relevant for analyzing the long run labor-market effects of a number of factors such as productivity growth, and tax or labor-market reforms. **JEL Codes**: E60, H20, J20. **Keywords**: CLM model; DLM model; Europe; hours worked per population; labor-market policy; long-run labor wedge; OECD countries; taxes; United States; U.S. tax puzzle.

^{*}The authors are thankful for very useful feedback provided during presentations at Bates College, Bentley University, the Federal Reserve Bank of Boston, Florida International University, IHEID, the Swiss Society of Economics and Statistics Meetings, Texas State University, Trinity University, University of Colorado at Boulder, University of Houston, University of Miami, University of Nottingham, University of Texas at Dallas, University of Utah, and also for comments from David Arseneau, Michael Elsby, Julia Cajal, Mary Corcoran, Chris Foote, Jim Hines, Miguel Iraola, Brian Jacob, Taylor Jaworski, Miles Kimball, Omar Licandro, Rick Mansfield, Alessandro Peri, Celine Poilly, Alessandro Ruggieri, Manuel Santos, Matthew Shapiro, Joel Slemrod, Jeff Smith, Cédric Tille, and Ian Wright. All errors are our own.

[†]Corresponding author. Department of Economics. University of Massachusetts at Lowell. E-mail: epsteinb@gmail.com.

[‡]Department of Economics. Tufts University E-mail: Alan.Finkelstein Shapiro@tufts.edu.

[§]Department of International Economics, Graduate Institute of International and Development Studies, Maison de la Paix P1-619, Chemin Eugene Rigot 2, Geneva 1202, Switzerland. E-mail: rahul.mukherjee@graduateinstitute.ch.

[¶]Federal Reserve Bank of Chicago. E-mail: sramnath@frbchi.org.

1 Introduction

Hours worked per population are fundamentally important for aggregate economic activity. The contemporary canonical (dynamic, micro founded, representative agent) macroeconomic model is broadly established as a common benchmark that can get at the basic forces associated with the behavior of hours worked per population—which we denote by H—at business cycle frequency.¹. As is well known, this canonical model can explain some, but not all, of the short-run behavior of H. Yet, the existence of such a common framework has allowed the macroeconomics literature to succinctly identify a series of key factors that are critical—beyond these basic forces for successfully and with great tractability explaining the business cycle behavior of H within the representative-agent paradigm. These factors include, among others, search frictions and indivisible labor.²

The objective of this paper is to develop and propose a tractable benchmark model that can serve as a common reference for modeling the *trend/long-run* behavior of H. Absent such framework, it is not possible to identify the fundamental basic forces that are relevant for answering a series of timely questions of crucial importance and general interest. For instance, how should we expect trends in H to respond to the following developments: a slowdown in global output, which is a critical issue amid the ongoing convergence of the growth rate of the Chinese economy to that of advanced economies; changes in demographics, which is a critical issue in light of the aging of the population in advanced economies; economic policy, such as tax policy, labor market reforms, and pension reforms, which are critical in light of fairly recent policy developments in Europe in the wake of the Global Financial Crisis and amid the European Sovereign Debt Crisis.

In light of the above questions, the motivation behind our research objective is the following. In the macroeconomics literature, in contrast to the short-run behavior of H, an understanding of the *trend* behavior of H within the representative-agent

¹Of course, the benchmark reference for these models is Kydland and Prescott (1982).

²Diamond (1982) and Mortensen and Pissarides (1994) are key references regarding search frictions, and Hansen (1985) and Rogerson (1988) are so regarding indivisible labor.

paradigm is considerably limited. In particular, a common, widely accepted, and tractable benchmark model that can get at the basic forces associated with the trend behavior of H, and, critically, that can explain to a considerable degree the trend behavior of H, remains elusive.

Importantly, as is well known, for all purposes the canonical macroeconomic model can explain *none* of the *trend/long-run* behavior of H. This results in the model yielding a prominent long-run labor wedge, that is, the long-run difference between model-predicted H and empirical H.³ That said, a critical advancement in the understanding of the trend behavior of H in a representative-agent context took place with Prescott (2004), who proposed taxes as an intuitive variable that could help narrow the (long-run) labor wedge. However, Figure 1, which plots empirical H and modelpredicted H by tax-inclusive and -exclusive versions of the canonical model shows the following. While on average taxes narrow the labor wedge in Europe substantially, they dramatically widen the labor wedge in the United States. We refer to this as the "U.S. tax puzzle."⁴



Figure 1: Empirical hours worked per population and predictions from CLM model with and without taxes for Europe (top panel) and the US (bottom panel).⁵

³More technically, the (long-run) labor wedge captures the extent to which, in the long run, the marginal product of labor differs from the marginal rate of substitution of consumption for leisure.

⁴This result is endemic to this tax-related literature, which also includes, amont others, Ohanian et al. (2008) and McDaniel (2011).

⁵Notes: All data are at yearly frequency. Total work hours are from the Conference Board's

The model that we propose is an intuitive and tractable extension of the canonical macroeconomic model. Our model is Walrasian and representative-agent in nature, and nests the canonical model. Two key features distinguish our model from the canonical one. First, the presence of employment-attainment costs on the household side, which enter the model in the form of disutility. Second, the presence of firm-side employment adjustment costs.

While these two modeling ingredients are not novel from the conceptual standpoint, the technical way in which our model incorporates these ingredients is indeed novel. Critically, the way in which our model incorporates these two ingredients results in an intuitive and straightforward disentanglement of the two margins of labor hours worked per worker, which we denote by h, and the employment-population ratio, which we denote by e—on both the household and firm side. This is in contrast to the canonical macroeconomic model, where the only endogenous labor-market variable predicted by the model is H itself—we refer to this class of model as "combined labor margin (CLM) models." As such, our model's prediction of H is built from the bottom up, with the model predicting both h and e, the product of which is of course H. Therefore, we refer to our model as the "dual labor margins (DLM) model."

Clearly, there are many ways by which the two margins of labor can be disentangled. An advantage of our DLM model relative to alternative frameworks is that household-side employment-attainment costs allow our framework to remain within the representative-agent paradigm. Moreover, in contrast to labor search theory, where there is involuntary unemployment, our modeling methodology is such that any existing nonemployment is voluntary. Both of these features are critical for allowing our framework to nest the canonical macroeconomic model in straightforward fashion. Moreover, as noted below, the particular way in which our DLM model disentangles the two margins of labor allows it to speak directly to the U.S. tax puzzle.

Total Economy Database, employment is from the OECD, population data are from the UN, tax data are from McDaniel (2007), and consumption and output data are from the Penn Word Tables. "Europe" refers to the simple average of European countries in our sample. Sample countries, which are standard in related literature, and data span are limited by the availability of time series on taxes. Details are in the paper's data and theory sections.

In addition, the way in which our DLM framework incorporates firm-side employmentadjustment costs is such that it generates, in straightforward and intuitive fashion, the following two relationships: a positive and contemporaneous relationship between e and capital taxes and also between H and capital taxes, and no relationship between h and capital taxes. Importantly, we document that these relationships exist in the data—a set of novel facts—which lends validity to our framework. Moreover, while the empirical positive relationship between e and capital taxes is in principle puzzling, our DLM model implies that this relationship is not causal. It is instead an observed outcome stemming from forward-looking optimal employment demand, and from this vantage point, in line with intuition, higher capital taxes put downward pressure on employment demand.

In line with related literature that studies the trend behavior of H, we operationalize our model using a business cycle accounting approach.⁶ We show that our model can account for the trend behavior of h (the equilibrium condition for which is static in our model) and e (the equilibrium condition for which is a novel *dynamic* expression) very well in *both* Europe *and the United States*. Therefore, our model can account very well for the trend behavior of H across countries. As such, our model is an important step forward in understanding the trend behavior of H relative to the canonical macroeconomic model and, therefore, in addressing the U.S. tax puzzle.

Our work results in the following three contributions. First, our DLM model itself. Indeed, the extent to which our DLM framework is successful in explaining the trend behavior of h, e, and therefore H across countries suggests that our model can justifiably be proposed as a benchmark for analyzing the trend behavior of these variables. Second, we further validate the role that taxes play as related to the trend behavior of H as highlighted by earlier literature starting with Prescott (2004), even though an endemic feature of this literature is the existence of the U.S. tax puzzle In particular and for instance, while our DLM model highlights that productivity is a major driving force associated with the trend behavior of H in the United States—

 $^{^6\}mathrm{Key}$ references are, for example, Prescott (2004) and Chari, Kehoe, and McGrattan (2007) among others.

which is something not highlighted by earlier directly-related literature—taxes are more important for long-run contours in the trend behavior of H. Moreover, as related to taxes, our DLM framework also shows that a positive relationship between e and capital taxes and H and capital taxes observed in the data (a novel stylized fact that we document) has no causal implications. Instead, these relationships are an expost outcome stemming from capital taxes putting downward pressure—in line with intuition—on forward-looking employment demand. This has not been highlighted by earlier literature.

Third, since our DLM model does very well in explaining the trend behavior of H in the United States, our DLM model thoroughly contributes to resolving the U.S. tax puzzle, for which our results imply that disentangling the two margins of labor is crucial. Of note, our DLM model's equation for h, which is rigorously micro founded, is *exactly the same* as the canonical model's equation for H. Our DLM results therefore imply that in terms of trends, the canonical model will give the *impression* of correctly predicting trends in H if these trends are driven by h (as we review later in the paper, this is the case empirically on average in Europe, where as noted above the tax-inclusive canonical model is successful—*in appearance*). In contrast, the canonical model will give the *impression* of failing to predict trends in H if these trends are driven by trends in e (as we review later in the paper, this is the case empirically on of failing to predict trends in H if these trends are driven by trends in e (as we review later in the paper, this is the case empirically in the United States, where as noted above the canonical model predicts *counterfactual* H). Ultimately, this implies that a large fraction of the long-run labor wedge that remains after accounting for taxes is employment itself.

Of course, it is well known that the canonical macroeconomic model faces important limitations in accounting for the behavior of H at business cycle frequency given an underlying lack of explanatory power over the behavior of e (hence, for instance and as noted earlier, the relevance of labor search and indivisible labor theories). However, that this also be the case at the trend level is surprising. Indeed, to the extent that the macroeconomic conceptual framework for modeling the long run is correct, then, with perfectly competitive markets and fully variable inputs, why would the long-run behavior of H depend notably on anything else except the long-run behavior of the marginal rate of substitution of consumption for leisure and the long-run behavior of the marginal product of labor?

This paper proceeds as follows. Section 2 reviews related literature. Section 3 details the data we use in our analysis. Section 4 focuses on theory and Section 5 discusses its operationalization. Section 6 presents results. Section 7 concludes.

2 Related Literature

Our DLM framework begins as a full-fledged general equilibrium labor search model. This is so in order to thoroughly ground the key elements of our framework, which intuition-wise trace back to search theory. As such, we build on a vast related literature, including, among others, Diamond (1982), Mortensen and Pissarides (1994), and Merz (1995). That said, our baseline DLM specification is purged of all frictions that render the labor market noncompetitive, meaning that our baseline specification is Walrasian: all markets are competitive. Thus, our work is also related to existing studies that consider the extensive margin of labor in such frameworks, such as Hansen (1985), Rogerson (1987), Bils and Cho (1994), Cho and Cooley (1994), Mulligan (2001), Krusell et al. (2008), Llosa et al. (2014), and Erosa (2016), among others. Our DLM model contributes to these literatures by showing how accounting for household employment-attainment costs and firm-side employment adjustment costs are a means through which the extensive and intensive margins of labor can be easily disentangled in a representative-agent Walrasian framework.⁷

There is a large literature on the labor wedge—which is technically defined as the extent to which, in equilibrium, the marginal product of labor differs from the marginal rate of substitution of consumption for leisure. This literature includes, among others, Hall (2009), Shimer (2009), Pescatori and Tasci (2011), Karabarbounis (2014a and 2014b), Cheremukhin and Restrepo-Echavarria (2014), Cociuba and Ueberfeldt

⁷In our DLM framework, the way in which employment adjustment costs are introduced yields an intuitive relationship between employment and capital taxes. The broader relevance of adjustment costs is emphasized in a vast literature that includes, for instance, Cooper and Willis (2004), Caballero and Engel (2004), Cooper and Willis (2008), and Mumtaz and Zanetti (2014).

(2015), Gourio and Rudanko (2014), and Hou and Johri (2018). Focusing on hours worked per population, Prescott (2004) highlights that taxes are a natural candidate to help explain the *long-run* labor wedge.

Regarding the long-run labor wedge and taxes, Prescott (2004), Ohanian et al. (2008), and McDaniel (2011) are particular instances in which fairly standard taxinclusive versions of CLM models are used for the purposes of analysis. In all cases CLM models yield counterfactual predictions for U.S. hours worked per population (to greater or lesser degree), but successful predictions for hours worked per population for European countries. This means that even after accounting for taxes, the *long-run* U.S. labor wedge remains. Importantly, Ohanian et al. (2008) note explicitly that the discrepancy between empirical hours worked per population in the United States and hours predicted for the United States by CLM models are so stark that it is a crucial question for future research in macroeconomics. Yet, in very broad terms, the literature that studies the relationship between trends in taxes and trends in hours worked per population ends with McDaniel (2011). This impasse in the literature is perhaps the result of it being the case that obvious resolutions to the U.S. tax puzzle are not evident within CLM modeling frameworks.

We contribute broadly to the labor wedge literature beyond taxes by showing that in CLM frameworks a large portion of the *long-run* labor wedge stems from deficiencies in the ability of canonical models to account for the behavior of the extensive margin of labor. As noted in the Introduction, this deficiency is well known at business cycle frequency. However, as also noted in the Introduction, it is extremely surprising that this limitation is also present in the long run.

Importantly, our DLM model highlights that the U.S. tax puzzle can be successfully addressed within a Walrasian representative-agent modeling framework, that is, within the same framework in which the U.S. tax puzzle is originally observed. This stands in contrast with papers that highlight heterogeneity as critical for addressing the puzzle, such as Cociuba and Ueberfeldt (2015), who stress the role of gender and marital status. As such, our DLM model shows that heterogeneity is, in fact, not a critical factor for resolving the U.S. tax puzzle.

Finally, we highlight a literature that centers on the fact that, while a positive long-run relationship between productivity and equilibrium is intuitive and empirically relevant, a definitive theory that links these two variables is lacking. See, for instance, Layard, Nickell, and Jackman (1991), Blanchard (2007), who surveys the literature on traditional models of aggregate labor markets and concludes that in these models there is long-run neutrality of unemployment to productivity growth, Shimer (2010), and Elsby and Shapiro (2012). Complementing this literature our DLM model suggests that a direct link between changes in total factor productivity (TFP) and equilibrium employment can exist by TFP potentially affecting job-formation costs. Our structural estimation suggests that this is indeed the case and, moreover, that higher TFP lower job-formation costs, which establishes a long-run positive link between TFP and employment.

3 Data

Given our focus on the trend behavior of labor market variables, in line with related literature our analysis makes use of data at yearly frequency. The countries in our sample are those for which, in line with related literature, there is extensive time series data on taxes. These countries are the United States and the following eleven european countries: Austria, Belgium, Finland, France, Germany, Italy, Netherlands, Spain, Sweden, Switzerland, and the United Kingdom. Our analysis spans the years 1960 - 2014, since at the time of writing this paper those are the years over which tax data, which we need to operationalize our model, are available for these twelve countries—in line with related literature, our main tax data are average taxes from McDaniel (2007). These data are at yearly frequency and publicly available from the author's website.⁸

Regarding the additional data we need, following related literature we use data on consumption, output, and total factor productivity from the Penn World Tables (Feenstra et al., 2015), which are hosted by the Groningen Growth and Development

⁸http://www.caramcdaniel.com/

Center.⁹ Data on the working-age population (ages 15-64) come from the United Nations¹⁰, while data on aggregate work hours and aggregate employment come from the Conference Board's Total Economy Database.¹¹

Of note, the McDaniel tax series are *average* taxes (see the Appendix for details). For the United States, we are able to assess the sensitivity of results from using these data by comparing them to results obtained using a series on *average marginal* taxes available from the NBER.¹² To the best of our knowledge, there are no comparable tax series for the european countries in our sample. We show that results are similar in terms of trends, but the ability of our DLM model to track contours of the empirical data improves substantially. The main difference between the two labor tax series being that, as shown in Figure 3 below, the McDaniel series do not fully capture the impact of the Reagan tax reforms (1981 through 1986).¹³ This has a impact on the extent to which our DLM model can match the contour of H, but is fairly irrelevant for our DLM model's ability to match the trend behavior of H.

Table 1 presents the notation we use for empirical data through the remainder of paper. In turn, Figures 2, 3, and 4 show the behavior of these data over our sample period. For brevity, in the main text we present *graphical results* for the United States and Europe, only, where "Europe" refers to the simple average of the eleven european

⁹These data are available for download at https://www.rug.nl/ggdc/productivity/pwt/. Regarding TFP, these data feature, for each country, a TFP index, which is TFP relative to the United States in each period. Of course, then, U.S. TFP is normalized to 1 in all periods. Therefore, we construct U.S. TFP using a standard Solow residual approach and then use the Penn World Tables TFP indices to back out implied TFP levels for all other countries in our sample.

¹⁰These data are publicly available at https://esa.un.org/unpd/wpp/Download/Standard/Population/

¹¹These data can be found at https://www.conference-board.org/data/economydatabase/ While these data are publicly available, at the time of writing this paper accessing the data requires creating an account, which is free of charge.

¹²Tables 1 and 3 from http://users.nber.org/~taxsim/marginal-tax-rates/

¹³NBER tax series is available for a bit shorter time horizon.

countries in our sample.¹⁴

	1
Notation	Meaning
C/Y	Consumption-output ratio
1 + C tax	Return-adjusted consumption tax, $1 - \tau^c$ (McDaniel, 2007)
1 - L tax	Return-adjusted labor tax, $1 - \tau^l$ (McDaniel, 2007)
1 - K tax	Return-adjusted capital tax, $1 - \tau^k$ (McDaniel, 2007)
$1 - NBER \ L \ tax$	Return-adjusted NBER labor tax
TFP	Total factor productivity
$d\ln\left(x ight)$	Growth rate of variable x
$H \ empirical$	Hours worked per population
H fix h	H holding h fixed at 1960 level $(h_{1960} \cdot e_t)$
H fix e	H holding e fixed at 1960 level $(h_t \cdot e_{1960})$
$h \ empirical$	Hours worked per worker
$e \ empirical$	Employment-population ratio

 Table 1: Notation for Empirical Variables

Some highlights follow. First, Figure 2 shows that over the last several decades the main driving force behind the behavior of H in the United states is e, with hbeing relatively flat.¹⁵ The exact opposite is true of Europe, but with the behavior of e being virtually flat in absolute terms. Figure 3 shows that in the United States C/Y rose, τ^k decreased, τ^c was fairly flat, and τ^l rose. In Europe, C/Y, decreased, τ^k rose, τ^c rose somewhat as well, and there was a large secular increase in τ^l . Finally, Figure 4 shows that that TFP in the United States and Europe has followed a similar secular increase, with TFP growth slowing in Europe only towards the end of our sample (in this figure Europe's TFP in 1960 is relative to that of the United States, which is normalized to 100).

¹⁴On average the patterns for Europe reflect those across the European countries in our sample, so no single country drives the average. Also, results are robust to the averaging methodology including, for instance, GDP-weighted averages.

¹⁵Rogerson (2006) and Blundell et al. (2011) also broadly highlight these facts.



Figure 2: Hours worked per population, per worker, and the employment-population ratio in Europe (bottom panel) and the US (top panel).¹⁶



Figure 3: Empirical macroeconomic series for Europe (bottom panel) and the US (top panel).¹⁷

¹⁶Note: total work hours are from the Conference Board's Total Economy Database, employment is from the OECD, and population data are from the UN.

¹⁷Notes: data on total work hours are from the Conference Board's Total Economy Database, employment is from the OECD, population data are from the UN, tax data from McDaniel (2007), and output and consumption from the Penn Word Tables.





Table 2 shows results from running a panel regression with all countries in our sample, where the left-hand side variable is, alternatively, $d \ln (e_t)$, $d \ln (h_t)$, and $d \ln (H_t)$. The regressands are the contemporaneous growth rates of return-adjusted taxes (τ^i is the investment tax). For each left-hand side variable this regression is run with and without a business cycle control: the growth rate of output per population, $d \ln (Y_t)$. Consumption taxes are insignificant, labor taxes have the correct sign (higher τ^l puts downward pressure on labor market variables) throughout and are significant on net, and investment taxes are not significant.

Finally, note that capital taxes are contemporaneously positively and significantly associated with e and H, but not with h. To the best of our knowledge, this is a new stylized fact, which implies that the association of H with capital taxes seeps in via the association of e with capital taxes. As shown below, our DLM model suggests that this contemporaneous positive relationship between capital taxes and e and H is not causal. Instead, our DLM model implies that these relationships should be observed as an ex-post outcome of forward looking employment demand from previous periods, with, in line with intuition, expected increases in capital taxes putting downward pressure on employment demand (we elaborate on this relationship later).

¹⁸Notes: data are from the Penn World tables.

Variable	$d\ln(e_t)$	$d\ln(e_t)$	$d\ln(h_t)$	$d\ln(h_t)$	$d\ln(H_t)$	$\frac{1}{d\ln(H_t)}$
$d\ln(1+\tau_t^c)$	0.02	-0.33	0.08	0.01	0.10	-0.32
	(0.61)	(0.34)	(0.13)	(0.08)	(0.71)	(0.37)
$d\ln(1-\tau_t^l)$	0.22^{**}	0.16^{**}	0.07^{*}	0.05	0.29^{***}	0.21^{***}
	(0.09)	(0.07)	(0.03)	(0.03)	(0.07)	(0.06)
$d\ln(1-\tau_t^k)$	-0.18^{**}	-0.14^{**}	-0.05	-0.04	-0.23^{***}	-0.17^{***}
	(0.07)	(0.06)	(0.04)	(0.03)	(0.06)	(0.04)
$d\ln(1+ au_t^i)$	0.01	0.59	-0.11	0.00	-0.10	0.59
	(1.09)	(0.53)	(0.22)	(0.11)	(1.29)	(0.62)
$d\ln(Y_t)$		0.29^{***}		0.06^{***}		0.35^{***}
		(0.04)		(0.01)		(0.04)
Country fix. eff.	yes	yes	yes	yes	yes	yes
Obs.	648	648	648	648	648	648

Table 2: Panel Regression of Labor Market Variables¹⁹

4 Theory

Our DLM model begins as a full-fledged general equilibrium labor search model. Among other benefits, this approach allows for a disciplined and well-grounded theoretical justification for the presence of household-side search effort and the way in which we incorporate firm-side employment adjustment costs. *However, the baseline specification of our DLM model is purged of certain elements of standard search theory. Amid this backdrop, in our baseline DLM model all markets are competitive, including the labor market.* The modeling decision to purge our bottom-line DLM specification of certain search elements is specifically guided by the principle of making our DLM model easily comparable to the CLM model and, moreover, making our DLM model a tractable and intuitive extension of the CLM model.

The aggregate population consists of a unit mass, and a household (*not social*) planner solves the household's optimization problem. In contrast to related literature, the economy's population is entirely selfish, atomistic, and autonomous. "Autonomous" in this paper means that each ("atomistic") household member has the power to renege on the household planner's solution if it is not incentive compatible.

¹⁹Notes: total work hours are from the Conference Board's Total Economy Database, employment is from the OECD, population data are from the UN, output is from the Penn World Tables, and tax data are from McDaniel (2007).

The resources of all economic agents that are in the household are pooled.

We assume that the household owns the economy's final-goods producing firm, and, without loss of generality, that the firm owns the economy's capital stock. Households can purchase corporate bonds and firms can issue debt. Corporate bonds and debt mature one period after being issued. Therefore, within any period the inflow of new bonds and debt is equivalent to the total stock of bonds and debt. The inclusion of debt guarantees that firms are able to pay any incurred vacancy-adjustment costs, which, as we show, ultimately link capital taxes to employment.

Mathematical details regarding all of the following can be found in the Appendix. Moving forward, all non-price variables are normalized by the aggregate population. Moreover, all price variables are normalized by the price of consumption.

4.1 The Household

In line with standard labor-search theory from the household's point of view employment evolves as follows:

$$e_{t} = (1 - \rho) e_{t-1} + F(\chi_{t}) p_{t} s_{t}.$$
(1)

Above e_t denotes employment, ρ is the exogenous job destruction probability, and s_t is the endogenous mass of job searchers. $F(\chi_t) p_t$ is the household's effective job finding probability. $p_t \in [0, 1]$ is exogenous, and χ_t is household-controlled search effort. $F' \geq 0$, $F'' \leq 0$, F(0) = 0, and $F_t \to 1$ as $\chi_t \to \infty$.²⁰ These diminishing returns to search effort are justified both theoretically and empirically (see, for instance, Pissarides, 2000, and Chirinko, 1984). All told, equation (1) says that period-t employment is equal to the sum of all individuals who were employed last period whose jobs were not destroyed, $(1 - \rho)e_{t-1}$, and the mass of successful contemporaneous searchers, $F(\chi_t) p_t s_t$.²¹ This equation is a constraint in the household's problem.

Also in line with standard labor-search theory we assume that *in equilibrium* all individuals participate in the labor market. Therefore, in equilibrium the mass of contemporary searchers is equal to the sum of all individuals who did not find a job

²⁰These assumptions guarantee that the effective job finding probability is never greater than 1, as should be the case.

 $^{^{21}}$ We follow the timing convention in Arseneau and Chugh (20120).

in the previous period, $1 - e_{t-1}$, and the mass of all individuals whose jobs were destroyed at the end of the previous period, ρe_{t-1} . As such,

$$s_t = (1 - e_{t-1}) + \rho e_{t-1}.$$
 (2)

Given the timing of the model in each period t there are three employment states. These states are newly employed workers (those who flow into employment in period t, $F(\chi_t) p_t s_t$), "old" employed workers (those whose jobs were not destroyed at the end of the previous period, $(1 - \rho)e_{t-1}$), and searchers who did not find jobs in this same period $(1 - F(\chi_t))p_t s_t$ —nonemployed individuals.

The household's lifetime utility \mathbf{U}_t is equal to the infinite sum of the weighted sum of the instantaneous utility of individuals in each employment state. As such,

$$\mathbf{U}_{t} \equiv \mathbb{E}_{t} \sum_{s=t}^{\infty} \beta^{s-t} \left\{ \begin{array}{c} \underbrace{\equiv v_{s}^{e,O}}_{s} \underbrace{\equiv v_{s}^{e,O}}_{s} \underbrace{\equiv v_{s}^{e,N}}_{s} \underbrace{= v_{s}^{e,N}}_{s} \underbrace{= v_{s}^{old} \left[u(C_{s}^{e,old}) - G(h_{s}^{old}) \right]}_{s} + e_{s}^{new} \left[u(C_{s}^{e,new}) - G(h_{s}^{new}) - D\left(\chi_{t}\right) \right]}_{s} \right\} + \left(1 - e_{s}\right) \underbrace{[u(C_{s}^{n}) - D\left(\chi_{t}\right)]}_{s} \left[u(C_{s}^{n}) - D\left(\chi_{t}\right) \right]}$$

where β is the exogenous subjective discount factor, e_s^{old} denotes old employed individuals, e_s^{new} denotes new employed individuals, and $1 - e_s$ is the mass of period-*s* unsuccessful searchers (nonemployed individuals). $v_s^{e,old}$, $v_s^{e,new}$, and v_s^n are, respectively, the instantaneous utilities of each of these individuals, and $C_s^{e,old}$, $C_s^{e,new}$, and C_s^n denote the consumption these individuals.²² We assume that these consumptions are, respectively, a fraction $\kappa_s^{e,old}$, $\kappa_s^{e,new}$, and $1 - \kappa_s^{e,old} - \kappa_s^{e,new}$ of aggregate consumption C_s . Also, h_s^{old} is hours worked per old employed individual and h_s^{new} is hours worked per newly employed individual. Finally, u is utility from consumption (u' > 0 and u'' < 0), G is disutility from work hours (G' > 0 and G'' > 0), and D is disutility from search effort $(D' > 0, D'' > 0, \text{ and } D_t \to \infty \text{ as } \chi \to \infty$, where these last two assumptions are consistent with the assumptions on F).

As noted earlier, the economy's population is assumed to be atomistic, autonomous, and selfish. Therefore the household planner faces a series of *incentive compatibility*

²²The instantaneous utility of individuals in old employment does not include disutility from search effort since they do not search in the current period. In contrast, both newly employed individuals and unsuccesful searchers expend contemporaneous search effort, which of course is the same in equilibrium.

(participation) constraints. First, $v_t^{e,old} \ge v_t^{e,new}$. Second, $v_t^{e,new} \ge v_t^n$. Third, $v_t^n \ge \bar{v}_t$, where \bar{v}_t denotes the outside option of individuals from reneging on the household planner's decision and leaving the household. Jointly, these constraints guarantee that individuals will accept the household planner's proposed solution and pool their resources within the household.

The household's *budget constraint* is

$$(1 + \tau_t^c) C_t + (b_t - b_{t-1}) \le (1 - \tau_t^l) w_t (h_t^{old} e_t^{old} + h_t^{new} e_t^{new}) + UB \cdot s_t + V_t + (1 - \tau_t^k) r_{t-1} b_{t-1} + T_t.$$
(3)

Above, τ_t^c is the consumption tax, b_t denotes period-t bonds, τ_t^l is the labor tax, w_t is the real wage, UB denotes unemployment benefits, which are paid to every individual who searches in a period and is unsuccessful in finding a job, V_t denotes net-of-(capital)-tax dividends paid by the firm to the household (recall that the household owns the firm), τ_t^k is the capital tax, r_t is the period-t real interest rate, and T_t denotes government transfers. The household takes all taxes, prices, the unemployment benefit, dividends, and transfers as given.

The household planner's choice variables are the following. C_t , $\kappa_t^{e,old}$, $\kappa_t^{e,new}$, κ_t^n , h_t^{old} , h_t^{new} , e_t , s_t , χ_t , and b_t . Of course, given that $e_t^{old} = (1 - \rho) e_{t-1}$, then knowing s_t and e_t is sufficient to know the distribution of the entire population across employment states.

In what follows, for brevity, we focus on first order conditions relevant for the labor market, only. In addition, we present the household's optimality conditions for the baseline version of our DLM model, which is purged from certain elements of labor search theory (optimality conditions from the full-fledged search model are in the Appendix). In particular, our purging assumes: $p_t = 1$ in all periods t, $\rho = 1$, and UB = 0. These assumptions are broadly in line with those used in Arseneau and Chugh (2012) when the authors show how to collapse their general equilibrium labor search model to a standard real business cycle (RBC) model. Given these assumptions equation (1) collapses to $e_t = F(\chi_t)$.

As shown in the Appendix, in equilibrium $h_t^{new} = h_t^{old} = h_t$, which is intuitive.

As also shown in the Appendix, as a result of the incentive compatibility constraints, the planner's optimal choice of $\kappa_t^{e,old}$ and $\kappa_t^{e,new}$ implies that in equilibrium in each period t the instantaneous utility of individuals in each employment state is equalized. Therefore, in equilibrium individuals are indifferent between being employed or nonemployed, so there is no involuntary unemployment. This is the case regardless of the presence of search frictions, and that is why to make this clear we refer to unsuccessful searchers as "nonemployed" instead of "unemployed."

The remaining relevant optimality conditions are as follows. For aggregate consumption: $u'(\kappa_t^{e,new}C_t)\kappa_t^{e,new} = (1 + \tau_t^c)\lambda_t$. This is entirely standard in tax-inclusive frameworks save for the presence of $\kappa_t^{e,new}$ that nonetheless does not affect the intuition behind this equation: the time-*t* effective (tax-inclusive) marginal utility of consumption is equal to the marginal value of real wealth, λ_t . For hours worked per worker

$$G'_t = \lambda_t \left(1 - \tau_t^l \right) w_t e_t, \tag{4}$$

meaning that the marginal cost of hours worked per worker equals its marginal benefit. Also, combining the optimality conditions for χ_t and e_t implies that

$$D'_t = \lambda_t \left(1 - \tau^l_t \right) w_t h_t \cdot F'_t.$$
(5)

Given our purging of labor search components, then in our baseline DLM specification choosing χ_t is the same as choosing e_t . As such, equation (5) is the household's effective employment supply equation, which means that the marginal cost of employment (the right-hand side of this equation, which is χ -dependent) is equal to its marginal benefit (the equation's left-hand side, which is also χ -dependent).

4.2 The Firm

Aggregate output Y_t is generated by the production function $Y_t = Y(Z_t, K_t, H_t)$. Here, Z_t is exogenous total factor productivity and K_t denotes capital. In line with standard CLM literature we assume that Y_t is linear in Z_t and increasing and concave in K_t and H_t . Of course, $H_t \equiv h_t \cdot e_t$. The firm's objective function, net-of-(capital)-tax dividends, is given by

$$\mathbb{E}_{t} \sum_{s=t}^{\infty} \Xi_{s|t} \left\{ \underbrace{\left(1 - \tau_{s}^{k} \right) \left[\begin{array}{c} Y\left(Z_{t}, K_{t}, h_{t}e_{t}\right) - w_{s}h_{s}e_{s} - I_{s} \\ -r_{s-1}d_{s-1} - \Lambda\left(Z_{s}\right) \cdot \left(\Phi v_{s} + \Omega\left(v_{s}/v_{s-1}, v_{s}w_{s}\right)\right) \right] \\ + \left(d_{s} - d_{s-1}\right) \\ = V_{s} \end{array} \right\}}_{\equiv V_{s}}.$$
(6)

Above, I_s is investment, d denotes debt, Φ is the exogenous flow cost of posting vacancies v_s . Ω is a standard adjustment cost function that is increasing and convex in the ratio v_s/v_{s-1} and also increasing $v_s w_s$. The product $v_s w_s$ captures in a reduced form way the intuition that hiring costs can reflect expenditures on a human resources department, and therefore are a fraction of the wage bill. Ω is equal to zero whenever v_s equals v_{s-1} , which, in particular, is the case in steady state (the broader relevance of adjustment costs for labor markets is emphasized by Cooper and Willis, 2004, Caballero and Engel, 2004, Cooper and Willis, 2008, and Mumtaz and Zanetti, 2014, among others.)²³ Turning to $\Lambda(Z_t)$, following, for instance, Pissarides (2000), we assume that the firm's costs of posting vacancies are *potentially* a function of aggregate productivity. However, we do not make any assumption on whether, if productivity is indeed related to vacancy posting costs, higher productivity makes it more costly or less costly for firms to post vacancies. Instead, as discussed below, we arrive at a conclusion regarding the potential relationship between productivity and vacancy posting via empirical analysis.

The firm faces two constraints. First, a standard equation of motion for the capital stock, $K_{t+1} = I_t + (1 - \delta) K_t$, where δ is the capital depreciation rate. Second, its perceived evolution of employment, $e_t = (1 - \rho) e_{t-1} + q_t v_t$, where q_t is the job filling probability, which the firm takes as given. In words, this equation says that the firm's contemporaneous stock of employed individuals is equal to the sum of all of

²³Of course, the change in the firm's debt position is not taxed, and, intuitively, we assume that $V_t \geq 0 \ \forall t$. Moreover, we do not include investment taxes since, per the evidence in Table 1 they do not have an impact on labor market variables and, furthermore, in our DLM model, in line with CLM models, investment taxes do not have a theoretical impact on labor market variables either.

the previous period's workers who did not lose their job, $(1 - \rho) e_{t-1}$, and all newly formed employment relationships, which are equal to the fraction of all vacancies posted that are filled in a period, $q_t v_t$.

The firm's choice variables are K_{t+1} (i.e., I_t), d_t , h_t , e_t , and v_t . For brevity, we present the firm's optimality condition for the baseline version of our DLM model, which, as noted earlier, is purged from certain elements of labor search theory (optimality conditions from the full-fledged search model are in the Appendix). In particular, as relevant for the firm, these purging assumptions are $q_t = 1$ in all periods t, $\rho = 1$, and $\Phi = 0$ (these assumptions are broadly in line with Arseneau and Chugh, 2012, when the authors show how to collapse their general equilibrium labor search model to a standard RBC model). In this environment the firm's equation of motion for employment collapses to $e_t = v_t$, so the firm's choice of vacancies is one and the same with its choice of employment.

We continue to focus on the labor market. The first order condition for h_t implies that

$$Y_{h_t} = w_t e_t,\tag{7}$$

which of course means that the marginal product of hours worked per worker is equal to its marginal cost and, therefore, that the wage is competitive. The first order condition for employment, which, recall, given our purging assumptions is one and the same with the first order condition for vacancies, is

$$-\Lambda\left(Z_{t}\right)\frac{\partial\Omega_{t}}{\partial e_{t}} = \mathbb{E}_{t}\Xi_{t+1|t}\frac{1-\tau_{t+1}^{k}}{1-\tau_{t}^{k}}\Lambda\left(Z_{t+1}\right)\frac{\partial\Omega_{t+1}}{\partial e_{t}}.$$
(8)

This means that whenever the firm is adjusting employment, employment demand is forward looking, as in standard models of labor demand with adjustment costs (see, for instance, Sargent, 1979).²⁴ Of note, the derivation of equation (8) uses the fact that from equation (7) $Y_{h,t}h_t = w_th_te_t$, which given the fact that $Y_{h_t}h_t$ equals $Y_{e_t}e_t$ implies, as a result, that $Y_{e,t}e_t = w_th_te_t$.

²⁴In periods in which the firm does not adjust employment demand, adjustment costs are zero and equation (8) implies that employment demand is pinned down by the following optimality condition: $Y_{e_t} = w_t h_t$.

Equation (8) means that firms find it optimal to equate the post-tax marginal cost of changing employment today to the post-tax (discounted) marginal cost of changing employment tomorrow. Another way to interpret this equation is that firms optimally decide on employment adjustment in order to smooth adjustment costs over time (and, more specifically, smooth the consumption value of net-of-(capital)-tax dividends for the household). As an example, suppose that at time t an expected increase in next period's capital tax rate τ_{t+1}^k lowers the ratio $\frac{1-\tau_{t+1}^k}{1-\tau_t^k}$. To restore equation (8) the firm will raise $\frac{\partial \Omega_t}{\partial e_t}$, which will require raising expected employment.

4.3 Closing the Model

Throughout the remainder of the paper we henceforth focus exclusively on the baseline ("purged") version of our DLM model. In particular, recall that this purging involves assuming the following: ρ , p_t , and q_t are equal to 1 in all periods, and UB and Φ are equal to zero in all periods. To close the model we assume that government consumption is zero, so that $T_t = \tau_t^c C_t + \tau_t^l w_t h_t e_t + \tau_t^k (r_{t-1}b_{t-1} + \bar{V}_t)$. This implies that the aggregate resource constraint is given by $Y_t = C_t + I_t + \Lambda(Z_t) \Omega(e_t/e_{t-1}, e_t w_t)$ whenever e_t is not equal to e_{t-1} , and $Y_t = C_t + I_t$ whenever the firm does not adjust employment. The model's equilibrium is discussed in the Appendix.

5 Operationalizing the Theory

Recall once more that as noted earlier throughout the remainder of the paper we focus exclusively on the "purged" version of our DLM model.

Following the literature most related to our paper, we evaluate our DLM model's performance by using the "business cycle accounting" approach.²⁵ For our purposes, this involves the following. Suppose that a variable X_t is a function of the vector of variables and parameters Ψ_t such that $X_t = X(\Psi_t)$. The model's performance regarding this variable is evaluated by taking the equation $X_t = X(\Psi_t)$ and feeding into it empirical data for Ψ_t , which results in a theory-implied prediction for the

 $^{^{25}\}mathrm{Key}$ references are, for example, Prescott (2004) and Chari, Kehoe, and McGrattan (2007) among others.

behavior of X_t . Assessing the model's fit then involves comparing this theory-implied prediction for the behavior of X_t with its empirical behavior.

Of course, our DLM model's equilibrium condition for h_t (equation (9)) is static, so it is straightforward to test this condition using business cycle accounting. That said, because employment demand (equation (8)) is dynamic, then so is equilibrium employment. Therefore, to be conceptually in line with business cycle accounting, we will assess the fit of our DLM model's dynamic equilibrium employment equation from an ex-post vantage point.

5.1 Functional Forms

In line with related literature we assume a standard constant returns to scale production function $Y_t = Z_t K_t^{\alpha} (h_t e_t)^{1-\alpha}$ where $\alpha \in (0, 1)$ and therefore $Y_{h_t} h_t = (1-\alpha) Y_t$ and $Y_{e_t} e_t = (1-\alpha) Y_t$. Turning to employment demand, we assume that $\Lambda (Z_t) = Z_t^{\zeta}$, where, as discussed earlier, ζ will be estimated and its value could, in principle, be less than zero, greater than zero, or equal to zero. We also assume the following cost function: $\Omega_t = \psi \mathbb{I}_t (e_t/e_{t-1})^{\phi} e_t w_t$, where $\psi > 0$, $\phi > 1$, and \mathbb{I}_t equals zero if e_t equals e_{t-1} and $\left(\frac{\beta\phi}{1+\phi}\right)^{t-1}$ otherwise. As such, \mathbb{I}_t guarantees that adjustment costs are zero whenever the firm is not adjusting vacancies (which, in particular, is the case in steady state), the fact that when adjusting employment \mathbb{I}_t equals $\left(\frac{\beta\phi}{1+\phi}\right)^{t-1}$ is a technical assumption that guarantees that the growth rate of employment is zero in steady state.

Turning to the household, following the assumptions in Shimer (2009) as applicable to the present context we assume $G(h_t) = \gamma \frac{\varepsilon}{1+\varepsilon} (h_t)^{\frac{1+\varepsilon}{\varepsilon}}$, where: γ and ε are parameters that are strictly greater than zero. As such, in our DLM model ε is the Frisch elasticity of the supply of hours worked per *worker*. Also following Shimer (2009), let $u(\cdot) =$ $\ln(\cdot)$. For expositional tractability we assume $F(\chi_t) = \varphi \chi_t^{\sigma}$, where $\sigma \in (0, 1)$ and $\varphi > 0$ (parameters are assumed to be in line with only infinite search effort being sufficient to approach a value of F_t equal to 1, even though our assumed functional form for F does not asymptote at 1), and also $D_t = \varrho \chi_t^{\theta}$, where $\theta > 1$ and $\varrho > 0$.

5.2 DLM Testable Implications

5.2.1 Hours Worked per Worker

As shown in the Appendix, combining the demand and supply of hours worked per worker implies the following equilibrium condition, which is one of two testable implications that our DLM model yields about the labor market:²⁶

$$h_t = \left[(1 - \alpha) \cdot \gamma^{-1} \left(1 - \tau_t^l \right) \cdot \left(1 + \tau_t^c \right)^{-1} C_t / Y_t \right]^{\frac{\varepsilon}{1 + \varepsilon}}.$$
(9)

Above, ε is the elasticity of hours worked per *worker* with respect to the wage. Intuitively, increases in labor taxes chip away at the value of the extra hour of work, increases in consumption taxes raise the price of consumption, and therefore increase the opportunity cost of consumption in terms of leisure, and a higher consumptionoutput ratio reduces the marginal value (in terms of consumption) of an additional hour of work.

To generate our DLM model's predictions of hours worked per worker, we proceed as follows. First, we run a panel of the regression corresponding to equation (9) after taking logarithms. Coefficients are constrained as implied by the theory, and we arrive at a single cross-country estimate of $\frac{\varepsilon}{1+\varepsilon}$, so for our purposes the preference parameter ε is the same across countries. Then, for each country in our sample, we feed countryspecific empirical time series data on consumption, consumption taxes, output, and labor taxes, into the right-hand side of equation (9) that, given our estimate of $\frac{\varepsilon}{1+\varepsilon}$, yield our DLM model's *country-specific* predictions for hours worked per worker. Since our interest is in long run trends, in line with related literature, such as Shimer (2009), the value of $\frac{1-\alpha}{\gamma}$ is chosen so that the average of predicted hours worked per worker match the average of their empirical counterpart on a country-by-country basis.

5.2.2 Employment

Turning to employment, with our assume functional forms and taking logs equation

$$C_t^{-1} = (1 + \tau_t^c) \,\lambda_t$$

 $^{^{26}\}mathrm{Given}$ our functional form assumptions first order condition for aggregate consumption is

Therefore, the optimal values of the κ variables are irrelevant for our analysis.

(8) yields the following "forward looking" condition for employment demand:²⁷

$$d\ln e_t = \frac{1+\phi}{\phi} d\ln e_{t+1} + \frac{1}{\phi} \begin{bmatrix} d\ln w_{t+1} + \zeta d\ln Z_{t+1} \\ -d\ln \lambda_{t+1} + d\ln \left(1 - \tau_{t+1}^k\right) \end{bmatrix}$$
(10)

(see the Appendix for details). Therefore, from the vantage point of the firm's period-t decision making equation (10) conveys the following optimal firm-side actions. Higher $d \ln w_{t+1}$, higher $d \ln e_{t+1}$, and lower $d \ln \lambda_{t+1}$ are associated with higher future output, given which the firm anticipates an expansion in future employment. In order to smooth adjustment costs, the firm frontloads some of this employment expansion, which puts upward pressure on $d \ln e_t$. For concreteness assume that $\zeta < 0$. As such, higher $d \ln Z_{t+1}$ means lower adjustment costs in the future. In smoothing these costs the firm postpones some contemporaneous adjustment, which puts downward pressure on $d \ln e_t$. Finally, higher capital taxes mean that future net-of-capital-tax dividends will be lower, given which the firm wants to adjust the least possible amount today in order to get as much net-of-tax-dividends today and, therefore, before the increase in capital taxes. This puts downward pressure on $d \ln e_t$.

It follows that the right-hand side of equation (10) reflects *causal* factors affecting the firm's contemporaneous demand. However, solving this equation for $d \ln e_{t+1}$ and lagging the equation one period gives us an ex-post perspective that, therefore, reflects *outcomes* (and *not contemporaneous causality*) that should be observed given the firm's decisions in the earlier period:

$$d\ln e_{t+1} = \frac{\phi}{1+\phi} d\ln e_t - \frac{1}{1+\phi} \begin{bmatrix} d\ln w_{t+1} + \zeta d\ln Z_{t+1} \\ -d\ln \lambda_{t+1} + d\ln \left(1 - \tau_{t+1}^k\right) \end{bmatrix}.$$
 (11)

Combining the dynamic equation for employment demand with a dynamic version of employment supply (see the Appendix for details) yields our DLM model's second testable implication, which is the following dynamic equation for equilibrium

²⁷As noted earlier, we will focus on from an ex-post perspective later in order to make business cycle accounting feasible, which is why here for simplicity we drop expectation operators.

employment:²⁸

$$d\ln e_t = \frac{\phi}{\phi + \left(\frac{\sigma}{\theta}\right)^{-1}} d\ln e_{t-1} + \frac{1}{\phi + \left(\frac{\sigma}{\theta}\right)^{-1}} \left[\begin{array}{c} \left(1 + \frac{\varepsilon}{1+\varepsilon}\right) d\ln \left(1 - \tau_t^l\right) - \frac{\varepsilon}{1+\varepsilon} d\ln \left(1 + \tau_t^c\right) \\ -\frac{\varepsilon}{1+\varepsilon} d\ln \left(C_t/Y_t\right) - d\ln \left(1 - \tau_t^k\right) - \zeta d\ln Z_t \end{array} \right].$$
(12)

Intuitively, this is an autoregressive moving average process with exogenous variables (ARMAX) for equilibrium employment—clearly, this equilibrium process is nonexplosive in employment. In this equation the variables scaled by coefficients that include ε trace back to the supply of hours worked per worker, of which employment supply is a function of. The terms that involve ε , which are $\left(1 + \frac{\varepsilon}{1+\varepsilon}\right) d \ln \left(1 - \tau_t^l\right)$, $\frac{\varepsilon}{1+\varepsilon} d \ln \left(1 + \tau_t^c\right)$, and $\frac{\varepsilon}{1+\varepsilon} d \ln (C_t/Y_t)$, also trace back to the supply of hours worked per worker, and their impact on equilibrium employment is akin to their impact on equilibrium h. In addition, the presence of σ/θ traces back to employment supply and reflects the relative degree of diminishing returns to search.²⁹

To operationalize equation (12) via business cycle accounting we first estimate its parameters. To be internally consistent, the regressors $\left(1 + \frac{\varepsilon}{1+\varepsilon}\right) d \ln \left(1 - \tau_t^l\right)$, $\frac{\varepsilon}{1+\varepsilon} d \ln \left(1 + \tau_t^c\right)$, and $\frac{\varepsilon}{1+\varepsilon} d \ln \left(C_t/Y_t\right)$ are generated using the estimate of ε obtained from from running the regression corresponding to equation (9). As such, our (constrained, as implied by the theory) regression of equation (12) yields estimates of $\frac{\phi}{\phi+\frac{\theta}{\sigma}}, \frac{1}{\phi+\frac{\theta}{\sigma}}, \text{ and } \frac{\zeta}{\phi+\frac{\theta}{\sigma}}.$

Recall that in estimating equation (9) we run a panel with all countries in our sample since with that equation we are ultimately estimating ε , which is a preference parameter and therefore should not be assumed to be different across countries. However, in estimating equation (12) we are guided by the fact that there are well-known

 $^{28}\mathrm{Moreover},$ the household's stochastic discount factor is

$$\Xi_{s|t} \equiv \beta^{s-t} C_t \cdot C_s^{-1} \left(1 + \tau_t^c\right) \left(1 + \tau_s^c\right)$$

where $s \ge t$, which means that higher consumption as well as higher consumption taxes (or, taken together, higher consumption expenditures) at time t lower the marginal value of consumption in that period.

²⁹It is straightforward to show that in steady state equilibrium employment is tax-wise only a function of labor taxes (decreasing) and consumption taxes (decreasing).

differences between the relative flexibility of European versus U.S. labor markets (see, for instance, Llosa et al., 2014). For example, consider the OECD's indicators of employment protection legislation (indices that go from 0, which implies least employment restrictions, to 6, which implies most employment restrictions)³⁰. In our European sample, the average index for: protection of permanent workers against individual and collective dismissal is 2.47; the average protection of permanent workers against individual dismissal is 2.16; specific requirements for collective dismissal is 3.24; and regulation on temporary forms of employment is 1.91. In contrast, in the United States these figures are, respectively: 1.17 (over 50 percent lower than in Europe); 0.49 (nearly 80 percent lower than in Europe); 2.88 (over 10 percent lower than in Europe); and 0.33 (over 80 percent lower than in Europe).

Importantly, note that in equation (12) all parameters to be estimated can be interpreted as reflecting, in a reduced form way, economic factors related to labor market rigidities. Indeed, a relatively higher value of ϕ means that it is more costly for firms to adjust employment, a relatively higher value of θ/σ means that relative returns to search decrease at a faster rate, and, assuming for concreteness that ζ is negative, a lower ζ means that higher productivity lessens employment adjustment costs by a relatively lower amount. In line with the evidence on labor market rigidities, it follows that there is no reason to expect that these parameters would be the same in Europe as in the United States. Thus, in estimating equation (12) we run a panel for Europe and a separate regression for the United States.

Given the estimated parameter values from running the regressions to estimate equation (12), we generate our DLM model-predicted level employment series for *each country* in our sample as follows. Let x_i denote empirical values for country *i*'s variable x, x_i denote predicted values for country *i* of variable x, and $\hat{\beta}$ denote the applicable vector of parameter estimates from running the regressions to estimate equation (12). First, we generate our DLM model-predicted employment growth rate for country *i* between the first growth rate that we can predict conditional on our

 $^{^{30}}$ https://www.oecd.org/employment/emp/oecdindicatorsofemploymentprotection.htm

data's starting point (period t-1 for the purposes of what follows immediately below before ending this section) using the following equation:

$$d\ln(e_{i,t}) = \hat{\boldsymbol{\beta}}' \times \begin{bmatrix} d\ln(e_{i,t-1}), d\ln(1-\tau_{i,t}^{l}), d\ln(1+\tau_{i,t}^{c}), \\ d\ln(C_{i,t}/Y), d\ln(1-\tau_{i,t}^{k}), d\ln(Z_{i,t}) \end{bmatrix}$$

Thereafter, we predict growth rates using for our lagged growth rates the model's predicted growth rates. In other words, the model's prediction of $d \ln (e_{i,t+1})$ uses $d \ln (e_{i,t})$. Therefore, except for the first lagged growth rate used, all other lagged growth rates used to get our model-generated predictions are fully and entirely endogenous.

Second, with our model-generated growth rates in hand, we then use the first empirical level of employment in period t along with the model-predicted growth rate between period t and period t + 1 to arrive at the model's first predicted level value of employment. We generate the remaining level values of employment by using the previous period's *predicted* employment levels. Akin to our DLM model's hours worked per worker predictions, the (level) employment series predicted by our DLM model is rescaled to have the same mean as the empirical employment series on a country-by-country basis.³¹

5.3 CLM Model H

We operationalize a representative version of the CLM model, in particular, the one used in Shimer (2009), in order to compare results from this model regarding H to the results from our DLM model. This version of the CLM model yields tax-inclusive results that are representative of related literature: a successful prediction of hours worked per population in Europe and highly counterfactual results for the United States.

In the CLM model the production function is the same as in our DLM model, and the household's instantaneous utility is $\ln(C_t) - \varepsilon (1 + \varepsilon)^{-1} (H_t)^{\frac{\varepsilon}{1+\varepsilon}}$, so in this case ε is the Frisch elasticity of the supply of hours worked per *population* (in contrast to

 $^{^{31}}$ We obtain our DLM model's predicted equilibrium employment series using its dynamic version since in our data no two adjacent employment figures are the same. In other words, in taking our model to the data the implication is that employment adjustment costs were paid in every period.

the DLM model, where ε is the Frisch elasticity of the supply of hours worked per *worker*). Of course, all taxes we consider in our DLM model remain the same in the CLM model, and in the CLM model the household's budget constraint is the same as in our DLM model.

All told, as shown in the Appendix, the following is the single equilibrium labor market condition that arises in the CLM model:

$$H_t = \left[(1-\alpha) \cdot \gamma^{-1} \left(1 - \tau_t^l \right) \cdot \left(1 + \tau_t^c \right)^{-1} C_t / Y_t \right]^{\frac{\varepsilon}{1+\varepsilon}}.$$
 (13)

(recall that the CLM model does not distinguish between the different margins of labor). Importantly, note that the right-hand side of equations (9) and (13) are the same (therefore, the intuition behind equation (13) is entirely analogous to the intuition behind equation (9)). Of note, this means that per our DLM model the CLM model's equation for equilibrium H is, in fact, an equation for equilibrium h.

In abstract terms it is straightforward to show that our DLM model's equilibrium condition for hours worked per worker is

$$h_t = G^{-1} \left(\lambda_t \left(1 - \tau_t^l \right) Y_{h,t} h_t \right)$$

and the CLM model's equilibrium condition for hours worked per population is

$$H_t = G^{-1} \left(\lambda_t \left(1 - \tau_t^l \right) Y_{H,t} H_t \right).$$

With perfect substitutability of e and h in production, then $Y_{h,t}h_t = Y_{H,t}H_t$ in the equations above. Therefore, the fact that our DLM model's equation for equilibrium h is the same as that of the CLM model for equilibrium H is not functional form dependent.

To operationalize equation (13) we take logarithms as in the case of the hours per worker equation from our DLM model and run a constrained panel regression to estimate the CLM model's value of ε . Then, we generate the CLM model-generated predicted H_t using the same methodology used to generate DLM-model predicted has discussed earlier. Results from this operationalization are those shown in Figure 1 in the paper's introduction.

6 Results

Recall that H is the only endogenous labor market variable predicted by the CLM model, and that the H predicted by our DLM model is built from the bottom up by first predicting h and e separately. Per the operationalization description in Section 5, our results obtain from predicting the CLM model's $H_{i,t}$ for each country i in our sample, and predicting our DLM model's $h_{i,t}$ and $e_{i,t}$ for each country in our sample and then constructing $H_{i,t}$ by multiplying these two. Model-generated results for Europe are the simple average across the model-predicted $H_{i,t}$ of european countries in our sample.

In what follows, we first show results from the regressions detailed earlier and highlight the implications of parameter estimates (Section 6.1). Then, we summarize model performance from a quantitative perspective across countries and on an individual basis (Section 6.2). In addition, we address model performance from the trend and contour perspective (Section 6.3). Finally, we focus on the implications of our model for the long-run labor wedge (Section 6.4).

6.1 Parameter Estimates

Table 3 shows results from running, in the fashion discussed earlier, constrained panel of the regressions corresponding to equations (9) and (13) in rows 1 and 2, respectively. All parameter estimates are statistically significant and of the correct sign. For our DLM model's h the implied estimate of ε is 1.44 (in this case the Frisch elasticity of the supply of hours worked per *worker*), and for the CLM model's H the implied estimate of ε is 2.84 (in this case the Frisch elasticity of hours worked per population).

Table 5. Estimates of Labor Endsticities					
Variable	$\ln(C_t/Y_t)$	$\ln(1+\tau_t^c)$	$\ln(1-\tau_t^l)$	Country fix. eff.	Obs.
$\ln(h_t)$	-0.59***	-0.59***	0.59^{***}	yes	660
	(0.02)	(0.02)	(0.02)		
$\ln(H_t)$	-0.74***	-0.74***	-0.74***	yes	660
	(0.03)	(0.03)	(0.03)		

Table 3: Estimates of Labor Elasticities³²

³²Notes: total work hours are from the Conference Board's Total Economy Database, employment is from the OECD, and population data are from the UN. Construction of model results use tax data from McDaniel (2007), and consumption and output data are from the Penn Word Tables.

Table 4 shows results from running the constrained regression of equation (12) in panel form for Europe (first row) and individually for the United States (second row). All parameter estimates are statistically significant. Moreover, the implied value of $\phi + \theta / \sigma$ is 11.11 in Europe and 4.16 in the United States. Give these values: (1.) the implied value ϕ is 5.56 in Europe and 1.78 in the United States, which means that the elasticity of employment adjustment costs with respect to the relative size of adjustment is over twice as high in Europe compared to the United States; and (2.) the implied value of θ/σ in Europe is 5.55 and 2.38 in the United States, which means that search returns decrease at a rate over twice as high in Europe compared to the United States. Finally, the implied value of ζ is -0.44 in Europe and -0.79 in the United States. Since ζ is negative, this means that when productivity is higher, employment adjustment costs are lower—intuitively, this can capture in a reduced form way that higher productivity can offset expansionary structural changes in the firm that must take place in order to accommodate more workers. Moreover, the fact that the value of ζ in the United States is less than its value in Europe implies that higher productivity decreases employment adjustment costs by less in Europe than the United States.

All told, we speculated that parameters in our DLM model's dynamic expression for equilibrium employment reflect underlying structural factors related to labor market rigidities. Results shown in Table 4 are indeed in line with this. Indeed, the implied rigidities from our results are consistent with these rigidities being substantially greater in Europe compared to the United States, which is in line with empirical evidence that studies these rigidities explicitly. *Taken together, these results lend substantial validity to our DLM model.*

Table 4: Estimates of Employment Equation $d \ln (e_t)$ regressand³³

	$d \ln$ of variable:							
Region	e_{t-1}	TFP_t	$\frac{C_t}{Y_t}$	$1 + \tau_t^c$	$d\ln 1 - \tau_t^k$	$d\ln 1 + \tau_t^l$	Country fix. eff.	Obs.
Europe	0.50^{***}	0.04^{**}	-0.09**	-0.09**	-0.09**	0.09**	yes	583
	(0.05)	(0.02)	(0.03)	(0.03)	(0.03)	(0.03)		
US	0.43^{**}	0.19^{**}	-0.24^{**}	-0.24^{**}	-0.24**	0.24^{**}		53
	(0.19)	(0.08)	(0.09)	(0.09)	(0.09)	(0.09)		

 $^{33}\mathrm{Notes:}$ total work hours are from the Conference Board's Total Economy Database, employment

6.2 Cross-Country Quantitative Performance

A key issue related to our research is understanding the extent to which our DLM model does relatively better or worse at predicting hours worked per *population* across countries compared to the CLM model. To get at this assessment, define for each country i in our sample the sum of squared deviations measure

$$SSD_{i} \equiv \sum_{t} \left(d \ln H_{i,t}^{DLM} - d \ln H_{i,t} \right)^{2} \cdot \left[\sum_{t} \left(d \ln H_{i,t}^{CLM} - d \ln H_{i,t} \right)^{2} \right]^{-1}$$

and the sum of absolute deviations measure

$$SAD_{i} \equiv \sum_{t} \left| d \ln H_{i,t}^{DLM} - d \ln H_{i,t} \right| \cdot \left[\sum_{t} \left(d \ln H_{i,t}^{CLM} - d \ln H_{i,t} \right) \right]^{-1},$$

where $d \ln H_{i,t}$ is the growth rate of empirical H and $d \ln H_{i,t}^{j}$ is model-predicted H by model $j \in \{CLM, DLM\}$. Then, for country i consider the numbers $100 \cdot (1 - SSD_{i})$ $\equiv \widetilde{SSD}_{i}$ and $100 \cdot (1 - SAD_{i}) \equiv \widetilde{SAD}_{i}$. This means that for $z_{i} \in \{\widetilde{SSD}_{i}, \widetilde{SAD}_{i}\}$ numbers $z_{i} > 0$ denote that our DLM does z_{i} percent better than the CLM model at predicting the growth rate of H, while numbers $z_{i} < 0$ denote that our DLM does z_{i} percent worse than the CLM model at predicting the growth rate of H. Of course, $z_{i} = 0$ implies that both models do equally well.

Table 5 shows \widetilde{SSD}_i and \widetilde{SAD}_i for all countries in our sample. On the basis of these metrics, we conclude that, compared to the CLM framework, our DLM model is an important step forward in understanding and predicting the behavior of hours worked per population. Results imply that on average our DLM model does better than the CLM model by between 12 and 20 percent, with notable improvements of up to 43 percent for the United Kingdom and 31 percent for the United States. The

is from the OECD, and population data are from the UN. Construction of model results use tax data from McDaniel (2007), and output, consumption and productivity data from the Penn Word Tables.

	iabie 9. eemp		Bin and D	nin model bei	Iormanoo	
(in %)	Austria	Belgium	Finland	France	Germany	Italy
\widetilde{SSD}_i	16.26	32.64	23.64	0.63	25.26	11.91
\widetilde{SAD}_i	4.21	26.5	7.89	-2.43	11.50	9.03
(in %)	Netherlands	Spain	Sweden	Switzerland	UK	US
\widetilde{SSD}_i	18.24	11.43	10.72	12.64	43.11	31.49
\widetilde{SAD}_i	10.62	7.83	8.19	7.69	28.15	22.7

only country for which model performance is about the same is France.

Table 5: Comparison of CLM and DLM model performance³⁴

6.3 Trend and Contour Analysis

Table 6 shows notation used for the purposes of presenting results in what follows. For our DLM model, Figure 5 shows results for h. Figure 6 shows results for e. Finally, Figure 7 shows results for H from both our DLM model and the CLM model.

-	
Notation	Meaning
h(e) DLM	Prediction per equation 9 (12)
H DLM	Product of h DLM and e DLM
h(e) DLM C/Y fix	Prediction per equation 9 (12)
	holding C/Y fixed
h(e) DLM cons tax fix	Prediction per equation 9 (12)
	holding τ_t^c fixed
h(e) DLM lab tax fix	Prediction per equation 9 (12)
	holding τ_t^l fixed
$e \ DLM \ cap \ tax \ fix$	Prediction per equation 9
	holding τ_t^k fixed
$e \ DLM \ TFP \ fix$	Prediction per equation 9
	holding TFP fixed

Table 6: Notation for DLM Model Generated Variables

Figure 5 shows that our DLM model tracks the behavior of h in both Europe and the United States very well. The same is true for e as shown in Figure 6. In each of these figures, counterfactuals ("fix") holding one at a time constant the ingredients that go into predictions of each variable reveal the most important driving forces associated with the behavior of each variable. Per Figure 5, in the United States, increases in C/Y and τ^l (recall Figure 3) have put downward pressure on h. In Europe the most important variable associated with the behavior of h is τ^l , with results implying that absent the substantial secular increase in τ^l (recall Figure 3) h

³⁴Notes: total work hours are from the Conference Board's Total Economy Database, employment is from the OECD, and population data are from the UN. Construction of model results use tax data from McDaniel (2007), and output, consumption and productivity data from the Penn Word Tables.

would have been flat.



<u>Figure 5:</u> Empirical hours worked per population and DLM model predictions for Europe (bottom panel) and the US (top panel).³⁵

Per Figure 6, in the United States, gains in TFP (recall Figure 4) are the most important factor associated with gains in employment, absent which e would have been flat. Taxes and C/Y play a secondary role, and mostly matter for the contour of e. That said, increases in C/Y and τ^l , and decreases in τ^k (recall Figure 3) jointly put downward pressure on e. (Recall that per our DLM model's equation for equilibrium employment, the contemporaneous relationship between capital taxes and employment is not causal, but instead an observed outcome of forward-looking employment demand from earlier periods given which, and in line with intuition, capital taxes put downward pressure on employment demand.) For Europe, the message regarding e from Figure 6 is quite stark: flat e is associated with the secular increase in τ^l offsetting gains in TFP that would have otherwise been associated with gains in e.

³⁵Notes: total work hours are from the Conference Board's Total Economy Database, employment is from the OECD, and population data are from the UN. Construction of model results use tax data from McDaniel (2007), and output and consumption data from the Penn Word Tables.



Figure 6: Empirical employment-population ratio and DLM model predictions for Europe (bottom panel) and the US (top panel).³⁶

Bringing the two margins of labor together, Figure 7 shows our DLM model's predictions for hours worked per population. Also shown are empirical H and, for reference, the CLM model's predictions as well (already shown in Figure 1). For Europe, both models perform successfully and fairly similarly. For the United States our DLM model matches the shallow V-shape of empirical H, while as discussed

³⁶Notes: employment is from the OECD, and population data are from the United Nations. Construction of model results use tax data from McDaniel (2007), and output, consumption and productivity data from the Penn Word Tables.



earlier the CLM model predicts a counterfactual secular decrease in H.

Figure 7 Empirical hours worked per population and DLM and CLM model predictions for Europe (bottom panel) and the US (top panel).³⁷

As highlighted earlier in the data section, the labor-tax series from McDaniel (2007), which are *average* taxes, do not reflect, contour wise, the Reagan tax reforms (1981 through 1986), which in contrast the NBER labor-tax series, which are *average* marginal taxes, indeed do reflect. Recall as well that both series have broadly the same trend, and that amid the Reagan tax reforms the McDaniel series imply rising labor taxes—which of course is driven by an increase in the tax base—while the NBER series imply decreasing taxes, as should be the case.

To assess the impact of these differences, Figure 8 shows results for e and H for the United States from operationalizing our DLM model with the NBER taxes, as well as results from the CLM model for H. Regarding e, comparing Figures 6 and 8 shows that our DLM model gets much better at the hump in employment starting in the early 1980s. This makes sense amid decreasing labor taxes as implied by the NBER tax series. Comparing Figures 7 and 8, the same is true for H. The endpoints of e and H predictions are much tighter as well. In addition, the CLM model's

³⁷Notes: total work hours are from the Conference Board's Total Economy Database, employment is from the OECD, and population data are from the United Nations. Construction of model results use tax data from McDaniel (2007), and output, consumption and productivity data from the Penn Word Tables.

predictions continue to yield counterfactual decreasing H.

In the Appendix we elaborate in further detail on the results described immediately above regarding the NBER taxes. We also show that an extension of our DLM model that accounted for differences in trends in employment by gender would for all purposes close the small remaining employment-hump gap between model and data that remains in the 1990s.



 $\frac{\text{Figure 8:}}{\text{taxes.}^{38}} \text{ Results for the United States using NBER labor taxes instead of McDaniel (2007) labor taxes.}$

6.4 Labor Wedge

Our DLM model implies that the CLM model's equation for H is in fact an equation for h. In turn, this implies that per our DLM model, after taxes are accounted for, an important fraction of the remaining labor wedge associated with the CLM model should be employment itself. To test this hypothesis, Figure 9 plots empirical H normalized to 1960, and "Hybrid CLM H" normalized to 1960 as well. These hybrid hours are the product of the tax-inclusive CLM model's predictions of H and *empirical e*. If the CLM model's H predictions are in fact predictions of h, then this

³⁸Notes: total work hours are from the Conference Board's Total Economy Database, employment is from the OECD, and population data are from the United Nations. Construction of model results use labor-tax data from the NBER, and output, consumption and productivity data from the Penn Word Tables.

product should track empirical H very well. As shown in Figure 9, this is indeed the case. This result lends further validity to our DLM model and, in particular, the implication that the CLM model's equation for H is in fact an equation for h and that after taxes are accounted for, in the CLM model a large fraction of the remaining labor wedge is employment itself.



Figure 9: Evidence regarding employment as the labor wedge.³⁹

Putting things in broader context, our results regarding the labor wedge imply that when hours worked per population are driven by hours worked per worker, as in the case of Europe (recall Figure 2), the CLM model can give the *impression, and impression only*, of correctly predicting hours worked per population, since the CLM model is in fact predicting hours worked per worker. In contrast, when empirical hours worked per population are driven by employment, as in the case of the United States (recall Figure 2) then the CLM model gives the *impression, and impression* only, of failing.

7 Conclusions

Hours worked per population (H) are fundamentally important for aggregate economic activity. The contemporary canonical macroeconomic model—which only

³⁹Notes: employment is from the OECD, and population data are from the United Nations. Construction of model results use tax data from McDaniel (2007), and output and consumption data from the Penn Word Tables.

yields predictions for H—gets at the basic behavior of this variable at business cycle frequency, but fails dramatically at the trend level. Importantly, a common, widely accepted, and tractable benchmark model that can get at the basic forces associated with the trend behavior of H, and, critically, that can explain to a considerable degree the trend behavior of H, remains elusive. Amid this backdrop, identifying the fundamental forces that matter for the response of H to a host of timely and critical trend-level developments is problematic. These developments include, but are not limited to, a slowdown in world output growth, changes in demographics that are currently critical for advanced economies, and policy issues such as labor-market reforms, pension reforms, and tax reforms.

In light of these facts, this paper develops and proposes a framework that can serve as a common reference for modeling the trend behavior of H: the DLM model. Our DLM model, which lies within the representative-agent Walrasian paradigm, is a novel but intuitive and tractable extension of the canonical model. In particular, our DLM model decomposes the trend-level determination of H into the two margins of labor hours worked per worker (h) and the employment-population ratio (e). Unlike earlier trend-level related literature centered on Walrasian representative agent contexts, our model does so via household-side employment attainment costs and firm-side employment adjustment costs.

Our DLM model identifies trends in the consumption-output ratio and taxes as important forces associated with the behavior of h. Our DLM model also explicitly identifies trends in total factor productivity as critically important for the trend behavior of e, which is something that by construction earlier related literature cannot get at, with taxes playing a second-order role for the United States—mattering mostly for the contour of e—but playing a more important role for Europe. Moreover, our DLM model can capture and explain a new stylized fact that we document, which is a cross-country relationship between capital taxes and e and H, but not with h. These relationships have not been highlighted by earlier related literature. On net, though, regarding taxes our model lends further validity to a literature that proposes taxes as an important factor that can help narrow the long-run labor wedge, even though an endemic feature of this literature is *stark counterfactual results for the United States*.

Our DLM model can track the behavior of h, e, and H very well in both the United States and a host of OECD countries. As such, our results suggest that our DLM model could indeed serve its intended purpose of being a framework that can serve as a common reference for modeling the trend behavior of H. This implies a broad and fruitful avenue for future research. In particular, in ongoing research we make use of our DLM model to explore the role of gender and age for the trend behavior of H.

References

- Arseneau, David M., and Sanjay K. Chugh. "Tax smoothing in frictional labor markets." *Journal of Political Economy* 120, no. 5 (2012): 926-985.
- [2] Bils, Mark, and Jang-Ok Cho. "Cyclical factor utilization." Journal of Monetary Economics 33, no. 2 (1994): 319-354.
- [3] Blanchard, Olivier. "A review of richard layard, stephen nickell, and richard jackman's Unemployment: Macroeconomic performance and the labour market." *Journal of Economic Literature* 45, no. 2 (2007): 410-418.
- [4] Blundell, Richard, Antoine Bozio, and Guy Laroque. "Labor supply and the extensive margin." *American Economic Review* 101, no. 3 (2011): 482-86.
- [5] Caballero, Ricardo J., and Eduardo MRA Engel. "A comment on the economics of labor adjustment: mind the gap: reply." *American Economic Review* 94, no. 4 (2004): 1238-1244.
- [6] Chari, Varadarajan V., Patrick J. Kehoe, and Ellen R. McGrattan. "Business cycle accounting." *Econometrica* 75, no. 3 (2007): 781-836.
- [7] Cheremukhin, Anton A., and Paulina Restrepo-Echavarria. "The labor wedge as a matching friction." *European Economic Review* 68 (2014): 71-92.
- [8] Chirinko, Robert S. "An empirical investigation of the returns to job search." *American Economic Review* 72, no. 3 (1982): 498-501.
- [9] Cho, Jang-Ok, and Thomas F. Cooley. "Employment and hours over the business cycle." *Journal of Economic Dynamics and Control* 18, no. 2 (1994): 411-432.
- [10] Cociuba, Simona E., and Alexander Ueberfeldt. "Heterogeneity and long-run changes in aggregate hours and the labor wedge." *Journal of Economic Dynamics* and Control 52 (2015): 75-95.
- [11] Cociuba, Simona E., Edward C. Prescott, and Alexander Ueberfeldt. "United States hours at work." *Economics Letters* 169 (2018): 87-90.
- [12] Cooper, Russell, and Jonathan L. Willis. "A comment on the economics of labor adjustment: Mind the gap." *American Economic Review* 94, no. 4 (2004): 1223-1237.

- [13] Cooper, Russell, and Jonathan L. Willis. "The cost of labor adjustment: Inferences from the gap." *Review of Economic Dynamics* 12, no. 4 (2009): 632-647.
- [14] Diamond, Peter A. "Wage determination and efficiency in search equilibrium." *The Review of Economic Studies* 49, no. 2 (1982): 217-227.
- [15] Elsby, Michael WL, and Matthew D. Shapiro. "Why does trend growth affect equilibrium employment? A new explanation of an old puzzle." *American Economic Review* 102, no. 4 (2012): 1378-1413.
- [16] Epstein, Brendan, Rahul Mukherjee, and Shanthi Ramnath. "Taxes and international risk sharing." Journal of International Economics 102 (2016): 310-326.
- [17] Erosa, Andrés, Luisa Fuster, and Gueorgui Kambourov. "Towards a microfounded theory of aggregate labour supply." *The Review of Economic Studies* 83, no. 3 (2016): 1001-1039.
- [18] Feenstra, Robert C., Robert Inklaar and Marcel P. Timmer. "The Next Generation of the Penn World Table." *American Economic Review* 105, no. 10 (2015): 3150-3182.
- [19] Gourio, Francois, and Leena Rudanko. "Can intangible capital explain cyclical movements in the labor wedge?." *American Economic Review* 104, no. 5 (2014): 183-88.
- [20] Hall, Robert E. "Reconciling cyclical movements in the marginal value of time and the marginal product of labor." *Journal of Political Economy* 117, no. 2 (2009): 281-323.
- [21] Hansen, Gary D. "Indivisible labor and the business cycle." Journal of Monetary Economics 16, no. 3 (1985): 309-327.
- [22] Hou, Keqiang, and Alok Johri. "Intangible capital, the labor wedge and the volatility of corporate profits." *Review of Economic Dynamics* 29 (2018): 216-234.
- [23] Karabarbounis, Loukas. "The labor wedge: MRS vs. MPN." Review of Economic Dynamics 17, no. 2 (2014a): 206-223.
- [24] Karabarbounis, Loukas. "Home production, labor wedges, and international business cycles." Journal of Monetary Economics 64 (2014b): 68-84.
- [25] Krusell, Per, Toshihiko Mukoyama, Richard Rogerson, and Ayşegül Şahin. "Aggregate implications of indivisible labor, incomplete markets, and labor market frictions." *Journal of Monetary Economics* 55, no. 5 (2008): 961-979.
- [26] Kydland, Finn E., and Edward C. Prescott. "Time to build and aggregate fluctuations." *Econometrica* (1982): 1345-1370.
- [27] Fang, Lei, and Cara McDaniel. "Home hours in the United States and Europe." The BE Journal of Macroeconomics 17, no. 1 (2016).
- [28] Layard, Richard, P. Richard G. Layard, Stephen J. Nickell, and Richard Jackman. Unemployment: macroeconomic performance and the labour market. Oxford University Press on Demand, 2005.

- [29] Llosa, Gonzalo, Lee Ohanian, and Andrea Raffo. "Firing Costs and Labor Market Fluctuations: A Cross-Country Analysis." *Mimeo* (2014).
- [30] McDaniel, Cara. "Average tax rates on consumption, investment, labor, and capital in the OECD 1950 2003." *Mimeo*.
- [31] Merz, Monika. "Search in the labor market and the real business cycle." *Journal* of Monetary Economics 36, no. 2 (1995): 269-300.
- [32] Mortensen, Dale T., and Christopher A. Pissarides. "Job creation and job destruction in the theory of unemployment." *The Review of Economic Studies* 61, no. 3 (1994): 397-415.
- [33] Mulligan, Casey B. "Aggregate implications of indivisible labor." Advances in Macroeconomics 1, no. 1 (2001).
- [34] Mumtaz, Haroon, and Francesco Zanetti. "Factor adjustment costs: A structural investigation." Journal of Economic Dynamics and Control 51 (2015): 341-355.
- [35] Ohanian, Lee, Andrea Raffo, and Richard Rogerson. "Long-term changes in labor supply and taxes: Evidence from OECD countries, 1956–2004." Journal of Monetary Economics 55, no. 8 (2008): 1353-1362.
- [36] Pescatori, Andrea, and Murat Tasci. Search frictions and the labor wedge. No. 11-117. International Monetary Fund, 2011.
- [37] Piketty, T. and Zucman, G., 2014. Capital is back: "Wealth-income ratios in rich countries 1700–2010." *Quarterly Journal of Economics*, 129(3), pp.1255-1310.
- [38] Pissarides, Christopher A. Equilibrium unemployment theory. MIT press, 2000.
- [39] Prescott, Edward C. "Why do Americans work so much more than Europeans?." No. w10316. National Bureau of Economic Research, 2004.
- [40] Rogerson, Richard. "Indivisible labor, lotteries and equilibrium." Journal of Monetary Economics 21, no. 1 (1988): 3-16.
- [41] Rogerson, Richard. "Understanding Differences in Hours Worked." Review of Economic Dynamics 9, (2006): 365-409.
- [42] Sargent, Thomas J. Macroeconomic theory. No. 339.2 S3. 1979.
- [43] Shimer, Robert. "Convergence in macroeconomics: The labor wedge." American Economic Journal: Macroeconomics 1, no. 1 (2009): 280-97.
- [44] Shimer, Robert. Labor Markets and Business Cycles. Princeton University Press, 2010.